Author’s response to reviews

Title: Cannabis exposure and risk of testicular cancer: a systematic review and meta-analysis

Authors:

Jason Gurney (jason.gurney@otago.ac.nz)
Caroline Shaw (caroline.shaw@otago.ac.nz)
James Stanley (james.stanley@otago.ac.nz)
Virginia Signal (virginia.signal@otago.ac.nz)
Diana Sarfati (diana.sarfati@otago.ac.nz)

Version: 2 Date: 02 Nov 2015

Author’s response to reviews:

3rd November, 2015

To Whom It May Concern,

Re: Revised version of manuscript investigating the association between cannabis use and development of testicular cancer

Thank you for the opportunity to present a revised version of our manuscript – titled “Cannabis exposure and risk of testicular cancer: a systematic review and meta-analysis” – for consideration for publication in BMC Cancer.

We have responded to each of the editor’s comments below, and have also attached a tracked-changes version of the manuscript:

1. Please replace the names of authors with their initials in the Authors' contributions section.

We have done this.

2. Please rename Introduction to Background.

We have done this.

3. Please include the missing email address on the title page of the manuscript.
We have done this.

4. Please include an Abbreviations section.

We have done this.

We look forward to your correspondence.

Best wishes,

Jason Gurney (on behalf of co-authors).